%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%  File          : ranlp-demo.tex
%%  Required Files: ranlp.sty &
%%                  ranlp.bst
%%                  epsf.tex         (they should be in the current directory)
%%  LaTeX version : LaTeX2e, 1994/12/01
%%  To Compile    : latex2e   ranlp-demo
%%                  bibtex    ranlp-demo
%%                  latex2e   ranlp-demo
%%                  latex2e   ranlp-demo
%%  To view       : xdvi ranlp-demo
%%  To generate PS file:
%%                  dvips -f ranlp-demo.dvi > ranlp-demo.ps
%%  To print      : lp -d <printer name> ranlp-demo.ps
%%
%%  Nicolas Nicolov <nicolas@watson.ibm.com>
%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\documentclass[11pt]{article}

\NeedsTeXFormat{LaTeX2e}[1994/12/01]    % Making sure that the right LaTeX is used

\usepackage{ranlp}
\usepackage{epsf}



\title{Sample Title: Centered, Bold; Size: 16pt}

\author{First Author\thanks{\ \ Start
\texttt{$\backslash$thanks\{...\}} with
``$\backslash$~$\backslash$~'' to get the space right between the
symbol and the begining of the first sentence. }
        \and
        Second Author\thanks{\ \ Thanks to the Uni for giving me a grant.}\\
        Author's Affiliation\\
    University of Arts and Science\\
        221 Baker Street\\
        Town ZIP, Country \\
    \texttt{\{email1,email2\}@dept.uni.ac.uk}}

\begin{document}

\maketitle

\begin{abstract}
In this paper we present the great idea that ... The word
'Abstract' should be centred, bold; 14pt. The body of the abstract
should be one paragraph and indented 6.25mm (0.25in) from the left
and right margin of the first column. The first sentence should
not be indented. The font size of the body of the abstract should
be smaller (10pt) or the same as the main text. If you use 12pt
for the main text you could use 10pt for the text of the abstract.
If the abstract is longer than half the textheight perhaps you are
trying to say too much. Do not provide a list of keywords! They
are useful for the reviewing stage mainly.
\end{abstract}


\section{Introduction}

This file contains the guidelines for camera-ready copies for the
proceedings of RANLP. The file contains some comments for people
using \LaTeX. If you don't use \LaTeX\ ignore these comments.

Main text:
The body of the text should be in double-column, single-spaced format;
size: 11pt (those authors that cannot use 11pt could use 10pt; If MS
Word users cannot use Computer Modern font they can use Times Roman 10pt).

The paper is organised as follows: ...


\section{\LaTeX\ users}

Relevant files for latex users (available on the web):

    ranlp.sty   style defining the layout of the paper
    ranlp.bst   bibliography style
    ranlp-demo.tex  simple demo file

\section{Page settings}

(1in = approx 2.54cm;
               A4 sheet is 210mm x 296mm / 8.25in x 11.625in)

Text height        = 247mm (9.6in).
White space between columns (Columnsep)
                   = 5mm (0.2in).
Textwidth          = 170mm (6.69in).
Width of columns   = (Textwidth - Columnsep)/2.
Indentation of the first lines of paragraphs
                   = 5mm (0.2in).
Left/Right margins = 2cm (0.78in)
   (the distance from the left/right edge of the paper to the left/right
    margin of the text)
                   =  (Width of A4 sheet - Textwidth)/2.
Top margin         = 2cm (0.78in).
Bottom margin      = 2.5cm (1in)
                   = Height of A4 sheet - Top margin - Text height.


\section{Section Titles}

1 Section Title         - bold; 14pt \\
1.1 Subsection Title        - bold; 12pt \\
1.1.1 Subsubection Title    - bold; 11pt (or 12pt) \\
Here are actual examples:

\subsection{Section Titles}

\subsubsection{Section Titles}

\noindent
If the section title is long and has to be on two (or more) lines then
indent the other lines are indented like this:

\section{A Rather Long Subsection w-w-w-w Title}

\section{Indentation}

No indentation of:
\begin{itemize}
\item   the paragraph after section headings;
\item   first paragraphs on new pages;
\item   first paragraphs of second column;
\item   paragraph after itemized environments (lists);
\item   paragraphs after figures/tables/equations.
\end{itemize}
\noindent
Use \texttt{$\backslash$noindent}.

\section{Figures/Tables}

Figures/Tables should be preferably within the text body and
appropriately enumerated. Under the contents of the Figure/Table there
should be a centered line which has the following format:

\centerline{Figure 2: Architecture of the system}

\noindent
If the figure/table is wide you may consider placing it so that it
spans the width of the two columns. In LaTeX one could use:

\begin{verbatim}
\begin{figure*}[...] ... \end{figure*}
\begin{table*}[...]  ... \end{table*}
\end{verbatim}

Referring to figures/tables:
The word Figure/Table should start with an initial capitals. Use:
``we show the control in Figure 3. The data is presented in Table~2.''

\begin{table*}[thb]
\centering\small
\begin{tabular}{lll}
% \hline
\textsc{Attribute 1} & \textsc{Attribute 2} & \textsc{Attribute 3} \\
\hline
item 1      & item 2    & 11\\
www ooo www & run       & 22\\
source      & preprocess    & 33\\
compile     & assemble  & 44\\
% \hline
\end{tabular}
\caption{Example table across the two columns}
\label{your-name:table:blah-blah}
\end{table*}

\section{Footnotes}

Length of the rule: 2cm (0.8in). Every footnote should be indented
0.5cm (0.2in). Font size of text - 10pt (or smaller if the size of
main body is 10pt).\footnote{Here is an example of a footnote. Put
minor comments or less central issues here. Avoid big footnotes though.}

\section{Citations}

Citation format for in line citations (in the body of the main text):
    (author-last-name year)
    (author-last-name \& author-last-name, year)
    (author-last-name et al. year)

Ex:     ... (Mellish 93)

Multiple citations should be separated with a semi-column and a space.

Ex:     ... (Clocksin \& Mellish 94; O'Keefe 90)
        ... (Gazdar \& Mellish 89)


\section{Length of Papers}


\begin{itemize}
\item   regular paper: no more than 8 pages
\item   short paper: no more than 7 pages
\item   poster: no more than 5 pages
\end{itemize}

\noindent
Do not worry if you have not used up the maximum number of pages.
Shorter, substantive papers are welcome!

\subsection{Oh, my paper is much longer}

That's normal. Most reviewers want me to elaborate on issues they
didn't understand. Obviously some bits will have to be omitted. But
I've spent so much time on these experiments! And this paragraph
looks so good! Don't worry --- everybody has the same problem.
Think about the main idea you want to convey. Maybe actually you are
trying to say more than one thing. Shouldn't the rest of the
important ideas be put in another paper. Maybe for next RANLP?
Then again I can have a nice well structured paper albeit a short one
which will interest people and they will contact me for the longer
report on which this paper is based. Maybe I can mention the URL for
that longer report or cite it.

\subsection{Cheating with the length}

Here are some obvious ideas, not all of them make my paper look very
nice but \ldots

\begin{small}
\begin{itemize}
\item   using \texttt{small} font for \texttt{itemize} environments
    and maybe some tables too;
    \emph{like here---see the source} \\[-1.5em]
\item   put this great but secondary idea in a footnote;\\[-1.5em]
\item   put a table/figure/equation in one column only;\\[-1.5em]
\item   redraw figures to take less space;\\[-1.5em]
\item   control vertical spacing between elements
    \texttt{$\backslash$vspace*\{-..\}} or
    \texttt{$\backslash\backslash$[-1.5em]};
    \emph{like here---see the source} \\[-1.5em]
\item   did I forget to put the bibliography in \texttt{scriptsize}
    (see Section~\ref{sec:Bibliography})?\\[-1.5em]
\item   Please don't touch margins or the text height!\\[-1.5em]
\end{itemize}
\end{small}


\section{Spellcheck the Source!}

Actually why don't I ask this friend of mine who is a native speaker
(who went to an exchange in the UK/Ireland/US/Canada/Australia) to have a
quick look at my paper. The spellchecker can't tell me if this phrase
sounds rather awkward in English. And it would be nice if the reader
flies over my text and not stumble on words and sentences trying to
find out what it is I was trying to say.

\section{Other}

Print on one side of the sheet only. On the reverse side of the sheet
please write (with a pencil) your name, title of paper, and number of page.
Please use a good quality laser printer.


************************

Send your submission electronically as a PDF file to RANLP2003 mail
(ranlp03@lml.bas.bg).

\noindent
Late camera-ready papers cannot not be included in the proceedings
(for obvious reasons). \textbf{Time is short!} Let us know if you are
encountering any problems.




\section{Bibliography}
\label{sec:Bibliography}

The word 'References:' should be left-aligned, bold; 14pt.
It should not be preceded by a number. The actual references should be
8pt (or 9pt if you have enough space), labeled and alphabetically
ordered according to the content of the labels. The label should be
the same as the in-line citation in the main text. All lines of a
single bibliographic reference except the first one should be indented
0.5cm (0.2in). Write the title of the paper/monograph in italics. Do
not underline it! If the authors are more than 2 please use the form:
(Shieber et al. 90) If you cite two different publications by the same
author which appeared in the same year use the form: (Nirenburg 94a)
(Nirenburg 94b), etc.

\LaTeX\ users:
\begin{verbatim}
    \bibliographystyle{ranlp}
    \begin{scriptsize}
    \bibliography{<YOUR_BIBTEX_FILE>}
    \end{scriptsize}
    \end{verbatim}

Ignore warning messages like:
\begin{small}
\begin{verbatim}
You can't pop an empty literal stack for entry
while executing---line 1275 of file ranlp.bst
\end{verbatim}
\end{small}

       when you bibtex your source. In your .bib file do not use:
           \texttt{AUTHOR = \{John Smith and others\},}
       because then the original style produces a wrong label:
       (Smith \& et al. 69). If there are two authors - specify the
       names of both authors. The label then looks like:
       (Smith \& Jones 69). If there are more than two authors specify
       the names of at least three authors. The label then looks like:
       (Smith et al. 69) as one would expect.


\section{Let's cite some refs and see what happens}

Here are some interesting references:
 \cite{Clocksin-Mellish'94},
 \cite{Dale-Mellish-Zock'90},
 \cite{Gazdar-Mellish'89},
 \cite{Mellish+'94},
 \cite{Nicolov-Mellish-Ritchie'95},
 \cite{Nicolov-Mellish-Ritchie'96},
 \cite{Noord'90},
 \cite{Reiter-Mellish'92},
 \cite{Mellish'93}.

You prpbably don't have the same bib file that is used with this document
but do have a look at the source (ranlp-demo.tex).

\bibliographystyle{ranlp}
\begin{scriptsize}
% Here below point to your bib file:
\bibliography{/home/nicolas/LaTeX/BibTeX/bib_data}
\end{scriptsize}

\end{document}
